Author's response to reviews

Title: The Polycythemia Vera revealed with bladder tumor in erectile disorder patient: a case report

Authors:

Ahmed Amine Bouchikhi (drbouchikhi@gmail.com)
MF Tazi (mft@lycos.com)
Diss Amiroune (d.amiroune@gmail.com)
JE Elammari (jea@hotmail.com)
A Khallouk (abKh@live.fr)
S Mellass (smel@lycos.com)
MJ Elfassi (elfa@live.fr)
MH Farih (far@hotmail.fr)

Version: 11 Date: 31 January 2013

Author's response to reviews: see over
Dear Editor of the Journal of Medical Case Reports,

Thank you for sending the four version of our corrected proof and suggesting correction that improved our paper.

The title from: "The Polycythemia Vera revealed in erectile disorder patient" should be to "The Polycythemia Vera revealed with bladder tumor in erectile disorder patient: a case report"

Thank you again and kind regards,
AA Bouchikhi,